Author’s response to reviews

Title: Exploration of possible association between Examiner stringency and Personality factors in clinical assessments: a pilot study

Authors:
Yvonne Finn (yvonne.finn@nuigalway.ie)
Peter Cantillon (peter.cantillon@nuigalway.ie)
Gerard Flaherty (gerflah@gmail.com)

Version: 2 Date: 23 April 2014

Author’s response to reviews: see over
Dear Aldrin and Editorial team BMC Medical Education,

I have received your email today with the request to add a title page at the front of our manuscript. This I have done and it includes the following:

- Title of manuscript
- Name, institution, email address and postal address of the corresponding author
- Name, institution and email addresses of the other 2 authors
- Short title of manuscript.

No other changes have been made to the manuscript.

Yours sincerely,

Dr Yvonne Finn
Lecturer in Clinical Methods and Clinical Practice, School of Medicine, National University of Ireland, Galway, Ireland